Author's response to reviews

Title: Teratoma of lumbosacral region- Site of extreme rarity: A case report

Authors:

mohd faheem Dr (faheemjnmc@gmail.com)
HASAN H SYED DR (hasanharris@rediffmail.com)
DINESH KARDAM DR (dinesh.kardam@rediffmail.com)
VEENA MAHESHWARI DR (vegovil@rediffmail.com)
ROOBINA KHAN DR (roobinakhan@rediffmail.com)
ATUL SHARMA DR (dr.atul111@gmail.com)

Version: 3 Date: 12 February 2011

Author's response to reviews: see over
Author's response to reviews
Title: TERATOMA OF LUMBOSACRAL REGION- SITE OF EXTREME RARITY: A CASE REPORT
Authors:
   1- MOHD. FAHEEM*(MF)- faheemjnmc@gmail.com
   2- SYED HASAN HARRIS(SHH)- hasanharris@rediffmail.com
   3- DINESH (DK) – dinesh.kardam@rediffmail.com
   4- VEENA MAHESHWARI(VM)- vegovil@rediffmail.com
   5- ROOBINA KHAN(RK)- roobinakhan@rediffmail.com
   6- ATUL SHARMA(AS)- dr.atul111@gmail.com

Version: 1 Date: 13 December 2010

Author's response to reviews: see over
Thank you for consideration of our manuscript for publication in your journal. We have reviewed the above manuscript according to your reviewer’s comments.

**Reviewer # 1 (Dr. Raheel Khan)**

Minor comment:

**Quality of written English:** Needs some language corrections before being Published

- Done

---

**Reviewer # 2 (Dr. Kavita Krishna)**

Minor comment:

The author should however discuss the possible differential diagnosis, of other midline tumours in this region.

- One whole paragraph on differential diagnosis of midline mass in lower back region has been added as suggested by the reviewer in discussion part of the case report.

Lumbosacral tumours - teratomas have been described in literature. So a review of some more case reports should also be mentioned.

- After extensive search of case report data base we are unable to find more references on the topic “lumbosacral teratoma in a **male** patient”